\begin{table}%T7
\caption{\label{tab:l1544}Fractional abundances in each model slab (with respect to~H$_2$), as well as total column densities and  overall deuterium fractionation predicted for L1544.}
%\centerline
{
\begin{tabular}{lcccccccc}
\hline\hline
          & 1& 2& 3& 4& 5& Total column & [XD]/[XH] &~Observed \\
                &&&&&&~density & &  col.~dens.\ \\
          &&&&&& (cm$^{-2}$) & & (cm$^{-2}$) \\
\hline
~$N$(H$_2$) (cm$^{-2}$) &4.5(+21) & 3.6(+21) & 8.7(+21) & 2.5(+22) & 7.6(+22) &  1.18(+23) & -- & 1.6(+23)$^a$\\
\hline
   CO      & 8.2($-$5)  & 4.4($-$5)  & 6.2($-$7) &  4.5($-$7)  & 2.2($-$13) &  5.5(+17) & -- & $\sim$9.0(+17)$^a$ \\ 
   ~O      & 1.8($-$4)  & 1.6($-$4)  & 9.0($-$8) &  9.8($-$8)  & 1.4($-$14) &  1.4(+18) & -- \\
 ~H$_3^+$  & 1.7($-$9)  & 7.3($-$10) & 1.1($-$9)  & 4.2($-$10) & 2.0($-$10) &  4.5(+13) & -- \\ 
~H$_2$D$^+$& 2.2($-$10) & 1.3($-$10) & 1.3($-$9)  & 6.7($-$10) & 3.1($-$10) &  5.4(+13) & 1.19 & 5.0(+13)$^c$\\ 
~HD$_2^+$  & 2.2($-$11) & 1.4($-$11) & 1.1($-$9)  & 7.0($-$10) & 3.6($-$10) &  5.4(+13) & 1.20 \\
 ~D$_3^+$  & 2.0($-$12) & 1.6($-$12) & 2.2($-$9)  & 4.9($-$9)  & 4.9($-$9)  &  5.2(+14) & 11.4 \\
~HCO$^+$   & 6.0($-$9)  & 3.3($-$9)  & 9.2($-$11) & 2.6($-$11) & 7.5($-$18) &  4.0(+13) & --  & 1.0(+14)$^d$\\ 
~DCO$^+$   & 3.3($-$10) & 2.2($-$10) & 1.4($-$10) & 1.3($-$10) & 7.7($-$17) &  6.8(+12) & 0.17 & 4.0(+12)$^d$\\ 
~N$_2$H$^+$& 3.1($-$10) & 2.2($-$10) & 4.9($-$10) & 6.9($-$12) & 1.4($-$12) &  6.7(+12) & -- & 2.0(+13)$^d$ \\ 
~N$_2$D$^+$& 1.5($-$11) & 1.4($-$11) & 6.7($-$10) & 3.3($-$11) & 1.3($-$11) &  7.8(+12) & 1.16 & 4.0(+12)$^d$ \\ 
 ~NH$_3$   & 3.2($-$8)  & 4.6($-$8)  & 6.1($-$8)  & 1.8($-$9)  & 3.9($-$10) &  9.2(+14) & -- \\ 
~NH$_2$D   & 2.1($-$10) & 5.7($-$10) & 3.2($-$8)  & 2.0($-$9)  & 1.2($-$9)  &  4.2(+14) & 0.46 \\
~NHD$_2$   & 6.7($-$13) & 3.3($-$12) & 8.3($-$9)  & 1.1($-$9)  & 1.1($-$9)  &  1.8(+14) & 0.20 \\
 ~ND$_3$   & 9.5($-$16) & 8.8($-$15) & 2.9($-$10) & 5.9($-$11) & 1.4($-$10) &  1.4(+13) & 0.02 \\ 
~H$_2$CO   & 7.5($-$9)  & 3.8($-$9)  & 2.1($-$9)  & 1.5($-$9)  & 1.8($-$19) &  1.0(+14) & --  & 2.5(+13)$^b$\\ 
~HDCO      & 4.9($-$10) & 2.7($-$10) & 7.4($-$10) & 7.4($-$10) & 3.3($-$19) &  2.8(+13) & 0.27 \\
~D$_2$CO   & 1.2($-$11) & 6.5($-$12) & 6.2($-$11) & 9.8($-$11) & 4.5e-20) &  3.1(+12) & 0.03 & 9.0(+11)$^b$ \\ 
 ~HCN      & 1.7($-$9)  & 1.2($-$9)  & 2.5($-$9)  & 1.4($-$10) & 3.3($-$17) &  3.8(+13) & -- \\ 
 ~DCN      & 4.5($-$11) & 4.6($-$11) & 9.0($-$10) & 3.0($-$10) & 3.1($-$16) &  1.6(+13) & 0.41 \\ 
 ~HNC      & 7.2($-$9)  & 5.4($-$9)  & 3.6($-$9)  & 3.1($-$10) & 1.5($-$16) &  9.0(+13) & --  & $\sim$9.0(+13)$^e$ \\ 
 ~DNC      & 5.2($-$11) & 5.5($-$11) & 1.6($-$9)  & 3.5($-$10) & 2.5($-$16) &  2.3(+13) & 0.26 & 3.0(+12)$^e$ \\ 
  ~NH      & 6.6($-$10) & 2.2($-$10) & 1.3($-$7)  & 3.0($-$9)  & 8.4($-$10) &  1.2(+15) & -- \\ 
  ~ND      & 2.3($-$11) & 1.0($-$11) & 1.6($-$7)  & 1.3($-$8)  & 7.9($-$9)  &  2.4(+15) & 1.92 \\ 
 ~NH$_2$   & 1.8($-$9)  & 6.5($-$10) & 3.4($-$8)  & 1.3($-$9)  & 4.0($-$10) &  3.7(+14) & -- \\ 
 ~NHD      & 7.0($-$12) & 5.1($-$12) & 1.2($-$8)  & 9.7($-$10) & 8.0($-$10) &  1.9(+14) & 0.52 \\ 
 ~ND$_2$   & 1.1($-$14) & 1.5($-$14) & 1.4($-$9)  & 2.0($-$10) & 2.7($-$10) &  3.7(+13) & 0.10 \\
   ~H      & 7.7($-$5)  & 2.8($-$5)  & 5.8($-$6)  & 2.0($-$6)  & 7.0($-$7)  &  6.0(+17) & -- \\ 
   ~D      & 5.1($-$7)  & 2.1($-$7)  & 9.2($-$7)  & 7.7($-$7) &  5.4($-$7)  &  7.1(+16) & 0.12 \\ 
 ~H$_2$O   & 6.2($-$7)  & 6.5($-$7)  & 6.9($-$9)  & 2.9($-$9)  & 4.2($-$16) &  5.3(+15) & -- \\ 
 ~HDO      & 1.5($-$8)  & 2.1($-$8)  & 3.0($-$9)  & 2.7($-$9)  & 7.1($-$16) &  2.4(+14) & 0.05 \\ 
 ~D$_2$O   & 3.3($-$11) & 6.0($-$11) & 4.8($-$10) & 1.0($-$9)  & 1.5($-$15) &  3.1(+13) & 0.01 \\
  CH$_4$   & 5.4($-$8)  & 2.0($-$8)  & 2.5($-$9)  & 1.4($-$9)  & 1.7($-$16) &  3.7(+14) & -- \\ 
 CH$_3$D   & 6.7($-$9)  & 2.6($-$9)  & 7.4($-$10) & 4.6($-$10) & 1.8($-$16) &  5.8(+13) & 0.16 \\ 
CH$_2$D$_2$& 4.4($-$10) & 1.8($-$10) & 5.9($-$11) & 4.7($-$11) & 5.1($-$17) &  4.3(+12) & 0.01 \\
  CH$_3$   & 6.0($-$11) & 9.9($-$12) & 5.0($-$9)  & 3.3($-$9)  & 3.6($-$15) &  1.3(+14) & -- \\ 
 CH$_2$D   & 5.6($-$12) & 9.6($-$13) & 1.3($-$9)  & 1.0($-$9)  & 2.6($-$15) &  3.7(+13) & 0.29 \\ 
CHD$_2$    & 2.7($-$13) & 4.9($-$14) & 9.1($-$11) & 1.1($-$10) & 1.2($-$15) &  3.6(+12) & 0.03 \\
 CD$_3$    & 4.1($-$15) & 8.3($-$16) & 1.2($-$12) & 1.5($-$12) & 1.1($-$17) &  4.8(+10) & 4($-$4) \\
 C$_2$H$_2$& 1.2($-$9)  & 8.7($-$10) & 9.7($-$10) & 5.0($-$10) & 3.3($-$16) &  3.0(+13) & -- \\ 
 C$_2$HD   & 6.2($-$11) & 4.5($-$11) & 5.9($-$10) & 6.8($-$10) & 9.8($-$16) &  2.3(+13) & 0.76 \\ 
  C$_4$H   & 3.0($-$12) & 4.7($-$13) & 7.7($-$10) & 4.8($-$10) & 3.5($-$15) &  1.9(+13) & -- \\ 
  C$_4$D   & 1.3($-$14) & 5.7($-$15) & 2.5($-$10) & 2.9($-$10) & 1.0($-$14) &  9.5(+12) & 0.51 \\ 
  C$_2$H   & 2.5($-$10) & 1.5($-$11) & 1.1($-$9)  & 5.4($-$10) & 5.4($-$16) &  2.4(+13) & -- \\ 
  C$_2$D   & 6.4($-$12) & 4.0($-$13) & 2.5($-$10) & 2.1($-$10) & 6.7($-$16) &  7.6(+12) & 0.31 \\ 
~H$_2$CS   & 7.5($-$13) & 1.4($-$13) & 7.9($-$12) & 5.0($-$12) & 5.1($-$18) &  2.0(+11) & -- \\ 
~HDCS      & 4.9($-$14) & 1.1($-$14) & 3.9($-$12) & 3.9($-$12) & 1.0($-$17) &  1.3(+11) & 0.67 \\ 
~D$_2$CS   & 9.8($-$16) & 2.4($-$16) & 3.6($-$13) & 6.0($-$13) & 2.7($-$18) &  1.8(+10) & 0.09 \\
\hline
\end{tabular}}
\smallskip
Note: Slab 1 is the most distant from the core centre.\\ 
 Refs.: $^a$Bacmann et~al.\ (\cite{Bacmann2002}), based on~$N$(C$^{17}$O$)\times1600$; $^b$Bacmann et~al.\ (\cite{Bacmann2003}); $^c$Caselli et~al.\ (\cite{Caselli2003}); $^d$Caselli et~al.\ (\cite{Caselli2002b}); $^e$Hirota et~al.\ (\cite{Hirota2003}), based on~$N$(HN$^{13}$C$)\times60$. 
\end{table}